Author's response to reviews

Title: Essential oil of pinus koraiensis leaves inhibits cell proliferation and migration via inhibition of p21-activated kinase 1 pathway in HCT116 colorectal cancer cells.

Authors:

Sun-Mi Cho (josunmi2000@hanmail.net)
Eun-Ok Lee (leook@khu.ac.kr)
Sung-Hoon Kim (sungkim7@khu.ac.kr)
Hyo-Jeong Lee (strong79@khu.ac.kr)

Version: 4 Date: 22 July 2014

Author's response to reviews: see over
Dear Dr. Tom Rowles, Executive Chief Editor:

We thank editor for valuable comments to our manuscript entitled with “Essential oil of pinus koraiensis leaves inhibits cell proliferation and migration via inhibition of p21-activated kinase 1 pathway in HCT116 colorectal cancer cells” (Manuscript ID: 1655723695125979)

Based on the Editorial comments, we revised them point by point as follows:

The followings are our point-by-point responses to reviewer’s comments

Editorial comments:

Please provide details in your Methods section on who undertook the formal identification of the plant material used in your study. Please also confirm whether a voucher specimen of this material has been deposited in a publicly available herbarium, and include this information in your manuscript. A deposition number should be included, if available.

(Response) Thank you for your comments. Your mentioned points were added in material and methods section.

We would like to take this opportunity to express our sincere thanks to the reviewers and editor who identified areas of our manuscript that needed corrections or modification for improving the quality of our manuscript. We hope our revised manuscripts would be accepted for publication in your journal soon.

Sincerely yours,

Hyo-Jeong Lee, Ph.D.